\documentclass[twocolumn,twoside]{article}
\makeatletter\if@twocolumn\PassOptionsToPackage{switch}{lineno}\else\fi\makeatother

\usepackage{amsfonts,amssymb,amsbsy,latexsym,amsmath,tabulary,graphicx,times,xcolor}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Following additional macros are required to function some 
% functions which are not available in the class used.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\usepackage{url,multirow,morefloats,floatflt,cancel,tfrupee}
\makeatletter


\AtBeginDocument{\@ifpackageloaded{textcomp}{}{\usepackage{textcomp}}}
\makeatother
\usepackage{colortbl}
\usepackage{xcolor}
\usepackage{pifont}
\usepackage[nointegrals]{wasysym}
\urlstyle{rm}
\makeatletter

%%%For Table column width calculation.
\def\mcWidth#1{\csname TY@F#1\endcsname+\tabcolsep}

%%Hacking center and right align for table
\def\cAlignHack{\rightskip\@flushglue\leftskip\@flushglue\parindent\z@\parfillskip\z@skip}
\def\rAlignHack{\rightskip\z@skip\leftskip\@flushglue \parindent\z@\parfillskip\z@skip}

%Etal definition in references
\@ifundefined{etal}{\def\etal{\textit{et~al}}}{}


%\if@twocolumn\usepackage{dblfloatfix}\fi
\usepackage{ifxetex}
\ifxetex\else\if@twocolumn\@ifpackageloaded{stfloats}{}{\usepackage{dblfloatfix}}\fi\fi

\AtBeginDocument{
\expandafter\ifx\csname eqalign\endcsname\relax
\def\eqalign#1{\null\vcenter{\def\\{\cr}\openup\jot\m@th
  \ialign{\strut$\displaystyle{##}$\hfil&$\displaystyle{{}##}$\hfil
      \crcr#1\crcr}}\,}
\fi
}

%For fixing hardfail when unicode letters appear inside table with endfloat
\AtBeginDocument{%
  \@ifpackageloaded{endfloat}%
   {\renewcommand\efloat@iwrite[1]{\immediate\expandafter\protected@write\csname efloat@post#1\endcsname{}}}{\newif\ifefloat@tables}%
}%

\def\BreakURLText#1{\@tfor\brk@tempa:=#1\do{\brk@tempa\hskip0pt}}
\let\lt=<
\let\gt=>
\def\processVert{\ifmmode|\else\textbar\fi}
\let\processvert\processVert

\@ifundefined{subparagraph}{
\def\subparagraph{\@startsection{paragraph}{5}{2\parindent}{0ex plus 0.1ex minus 0.1ex}%
{0ex}{\normalfont\small\itshape}}%
}{}

% These are now gobbled, so won't appear in the PDF.
\newcommand\role[1]{\unskip}
\newcommand\aucollab[1]{\unskip}
  
\@ifundefined{tsGraphicsScaleX}{\gdef\tsGraphicsScaleX{1}}{}
\@ifundefined{tsGraphicsScaleY}{\gdef\tsGraphicsScaleY{.9}}{}
% To automatically resize figures to fit inside the text area
\def\checkGraphicsWidth{\ifdim\Gin@nat@width>\linewidth
	\tsGraphicsScaleX\linewidth\else\Gin@nat@width\fi}

\def\checkGraphicsHeight{\ifdim\Gin@nat@height>.9\textheight
	\tsGraphicsScaleY\textheight\else\Gin@nat@height\fi}

\def\fixFloatSize#1{}%\@ifundefined{processdelayedfloats}{\setbox0=\hbox{\includegraphics{#1}}\ifnum\wd0<\columnwidth\relax\renewenvironment{figure*}{\begin{figure}}{\end{figure}}\fi}{}}
\let\ts@includegraphics\includegraphics

\def\inlinegraphic[#1]#2{{\edef\@tempa{#1}\edef\baseline@shift{\ifx\@tempa\@empty0\else#1\fi}\edef\tempZ{\the\numexpr(\numexpr(\baseline@shift*\f@size/100))}\protect\raisebox{\tempZ pt}{\ts@includegraphics{#2}}}}

%\renewcommand{\includegraphics}[1]{\ts@includegraphics[width=\checkGraphicsWidth]{#1}}
\AtBeginDocument{\def\includegraphics{\@ifnextchar[{\ts@includegraphics}{\ts@includegraphics[width=\checkGraphicsWidth,height=\checkGraphicsHeight,keepaspectratio]}}}

\DeclareMathAlphabet{\mathpzc}{OT1}{pzc}{m}{it}

\def\URL#1#2{\@ifundefined{href}{#2}{\href{#1}{#2}}}

%%For url break
\def\UrlOrds{\do\*\do\-\do\~\do\'\do\"\do\-}%
\g@addto@macro{\UrlBreaks}{\UrlOrds}



\edef\fntEncoding{\f@encoding}
\def\EUoneEnc{EU1}
\makeatother
\def\floatpagefraction{0.8} 
\def\dblfloatpagefraction{0.8}
\def\style#1#2{#2}
\def\xxxguillemotleft{\fontencoding{T1}\selectfont\guillemotleft}
\def\xxxguillemotright{\fontencoding{T1}\selectfont\guillemotright}

\newif\ifmultipleabstract\multipleabstractfalse%
\newenvironment{typesetAbstractGroup}{}{}%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\usepackage[authoryear]{natbib}



\newif\ifnokeywords\nokeywordsfalse

\makeatletter\input{size10-pointfive.clo}\makeatother%
\definecolor{kwdboxcolor}{RGB}{242,242,242}
\usepackage[hidelinks,colorlinks=true,allcolors=blue]{hyperref}
\linespread{1}                 
\def\floatpagefraction{0.8}

\usepackage[paperheight=11.69in,paperwidth=8.26in,top=1in,bottom=1in,left=1in,right=.75in,headsep=24pt]{geometry}

\usepackage{multirow-custom}

\makeatletter

\def\hlinewd#1{%
  \noalign{\ifnum0=`}\fi\hrule \@height #1%
  \futurelet\reserved@a\@xhline}
\def\tbltoprule{\hlinewd{1pt}\\[-14pt]}
\def\tblbottomrule{\noalign{\vspace*{6pt}}\hline\noalign{\vspace*{2pt}}}
\def\tblmidrule{\hline\noalign{\vspace*{2pt}}}



\let\@articleType\@empty
\let\@journalDoi\@empty
\let\@journalVolume\@empty
\let\@journalIssue\@empty
\let\@crossMarkLink\@empty
\let\@receivedDate\@empty
\let\@acceptedDate\@empty
\let\@revisedDate\@empty
\let\@publishedDate\@empty
\let\@copyrightYear\@empty
\let\@firstPage\@empty
\def\articleType#1{\gdef\@articleType{#1}}
\def\journalDoi#1{\gdef\@journalDoi{#1}}
\def\crossMarkLink#1{\gdef\@crossMarkLink{#1}}
\def\receivedDate#1{\gdef\@receivedDate{#1}}
\def\acceptedDate#1{\gdef\@acceptedDate{#1}}
\def\revisedDate#1{\gdef\@revisedDate{#1}}
\def\publishedDate#1{\gdef\@publishedDate{#1}}
\def\copyrightYear#1{\gdef\@copyrightYear{#1}}
\def\journalVolume#1{\gdef\@journalVolume{#1}}
\def\journalIssue#1{\gdef\@journalIssue{#1}}
\def\firstPage#1{\gdef\@firstPage{#1}}
\let\@issuelogo\@empty%
\let\@issuelogoex\@empty%
\def\issuelogo#1{\gdef\@issuelogo{#1}}
\def\issuelogoex#1{\gdef\@issuelogoex{#1}}


\def\author#1{%
  \gdef\@author{%
    \hskip-\dimexpr(\tabcolsep)\hskip5pt%
    \parbox{\dimexpr\textwidth-1pt}%
    {\fontsize{11}{13}\selectfont\raggedright #1}%
  }%
}


\usepackage{pharmascope-abs}
\usepackage{caption}
\usepackage{lastpage}
\usepackage{fancyhdr}
\usepackage[noindentafter,explicit]{titlesec}
\usepackage{fontspec}

\setmainfont[%
BoldFont=cambriab.ttf,%
ItalicFont=CAMBRIAI.TTF,%
BoldItalicFont=CAMBRIAZ.TTF]{Cambria.ttf}
\lefthyphenmin = 3

\def\title#1{%
  \gdef\@title{%
      \vspace*{-40pt}%
       \ifx\@articleType\@empty\else{\fontsize{10}{12}\scshape\selectfont\hspace{8pt}\@articleType\hfill\mbox{}\par\vspace{2pt}}\fi%
       \minipage{\linewidth}
         \hrulefill\\[-0.7pt]%%{\includegraphics[width=75pt,height=50pt]{ijrps_logo.png}}\hfill
         \mbox{~}\hspace{5pt}\parbox{.1\linewidth}{\IfFileExists{images/\@issuelogo.\@issuelogoex}{\includegraphics[width=79pt,height=61pt]{images/\@issuelogo.\@issuelogoex}}{\includegraphics[width=75pt,height=50pt]{ijrps_logo.png}}}\hfill
         \fcolorbox{kwdboxcolor}{kwdboxcolor}{\parbox{.792\linewidth}{%
          \begin{center}\fontsize{17}{17}\selectfont\scshape\vskip-7pt International Journal of Research in Pharmaceutical Sciences\hfill\end{center}%
          \vspace*{-10pt}\hspace*{4pt}{\fontsize{8}{9}\selectfont Published by JK Welfare \& Pharmascope Foundation\hfill  Journal Home Page: \href{https://ijrps.com}{\color{blue}\underline{\smash{https://ijrps.com}}}}\hspace*{4pt}\mbox{}}}%
       \par\vspace*{-1pt}\rule{\linewidth}{1.3pt}%
       \endminipage%
    \par\vspace*{9.2pt}\parbox{.98\linewidth}{\linespread{.9}\raggedright\fontsize{14}{17}\selectfont #1}%
    \vspace*{-8pt}%
  }
}


\setlength{\parindent}{0pt}
\setlength{\parskip}{0.4pc plus 1pt minus 1pt}

\def\abbrvJournalTitle{Int. J. Res. Pharm. Sci.}

\fancypagestyle{headings}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle,\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    ,\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[LO,RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ International Journal of Research in Pharmaceutical Sciences}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
}\pagestyle{headings}

                                                         
\fancypagestyle{plain}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}%
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle,\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    ,\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[LO,RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ International Journal of Research in Pharmaceutical Sciences}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
  \ifx\@firstPage\@empty\else\setcounter{page}{\@firstPage}\fi
}


\def\NormalBaseline{\def\baselinestretch{1.1}}
\usepackage{textcase}
\setcounter{secnumdepth}{0}

\titleformat{\section}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{13}\selectfont}
{\thesection}
{6pt}
{\MakeTextUppercase{#1}}
[]
\titleformat{\subsection}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsection}
{6pt}
{#1}
[]
\titleformat{\subsubsection}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsubsection}
{6pt}
{#1}
[]
\titleformat{\paragraph}[block]{\NormalBaseline\filright\fontsize{10.5}{10}\selectfont}
{\theparagraph}
{6pt}
{#1}
[]
\titleformat{\subparagraph}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubparagraph}
{6pt}
{#1}
[]

\titlespacing{\section}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsubsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\paragraph}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subparagraph}{0pt}{.5\baselineskip}{.5\baselineskip}

\captionsetup[figure]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}
\captionsetup[table]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}

\def\bibyear#1{#1}
%\def\bibjtitle#1{#1}   %%Publisher request
\def\bibauand{}
\setlength\bibsep{3pt}
\setlength\bibhang{8pt}

\makeatother
\date{}


\usepackage{float}

\begin{document}


\def\RunningAuthor{Souvik chakraborty et al.}
\firstPage{2612}
\articleType{Original Article}
\receivedDate{14.05.2019}
\acceptedDate{22.08.2019}
\revisedDate{17.08.2019}
\journalVolume{10}
\journalIssue{4}
\journalDoi{ijrps.v10i4.1519}
\copyrightYear{2019}
\def\authorCount{3}
\def\affCount{1}

\def\journalTitle{International Journal of Research in Pharmaceutical Sciences}

\title{Development of biodegradable scaffolds loaded with vancomycin micropartricles for the treatment of osteomyelitis}
\author{Souvik~chakraborty\textsuperscript{*},  
            Gowda~D~V, 
            Vishal~Gupta~N~\\[5pt]{Department of Pharmaceutics\unskip, JSS College of Pharmacy,  JSS Academy of Higher Education and Research, JSS Medical Institutions Campus, Shri Shivarathreeshwara Nagara, Mysuru-570015, Karnataka, India}}

\begin{abstract}
In this present research work, the development of biodegradable scaffolds loaded with Vancomycin micropartricles was carried out for the treatment of Osteomyelitis. Characterization Vancomycin Loaded microparticles and evaluation of the microparticles loaded scaffolds and also to carry out In-vitro release studies. Vancomycin hydrochloride microparticles were prepared with the help of double emulsion method. HPMC and Polaxomer 407 has been taken as the main polymers for the preparation of the microparticles. Chitosan was taken as the major polymer for the preparation of scaffolds for its greater biocompatibility and biodegradability. The preparation was done with the help of the solvent casting method. The formulation was taken for further characterization and evaluation studies. Fourier-Transform Infrared Spectroscopy and Differential scanning calorimetry were carried out for the pure vancomycin drug, and the chitosan polymer X-ray diffraction was carried out to check the crystallinity of the prepared scaffolds. The particle size, zeta potential and polydispersity index for vancomycin loaded microparticles were found to be 577.0\ensuremath{\pm}102.5 nm, 1624 mv and 0.254. The maximum and sustained release rate of the drug was found to be 95.6\ensuremath{\pm}0.478, at 16\ensuremath{^{th}} Hr. By taking all the reports, a conclusion can be drawn that, the formulated VLM biodegradable scaffolds will show burst release at the initial time of administration, which is essential for the wound healing activity and will be sustained throughout the process of treatment of osteomyelitis.
\end{abstract}\def\keywordstitle{Keywords}

\begin{keywords}Vancomycin hydrochloride,\newline Chitosan Polymer,\newline Microparticles,\newline Biodegradable Scaffolds
\end{keywords}
\twocolumn[ \maketitle {\printKwdAbsBox}]
\makeatletter{\raggedright\textsuperscript{*}Corresponding Author\par Name:\ Souvik~chakraborty~\\ Phone:\ +91 8095031565~\\  Email:\ souvik93pharmacist@gmail.com
        \par}\vspace*{-11pt}\hrulefill\par{\fontsize{12}{14}\selectfont ISSN: 0975-7538}\par% 
        \textsc{DOI:}\ \href{https://doi.org/10.26452/\@journalDoi}{\textcolor{blue}{\underline{\smash{https://doi.org/10.26452/\@journalDoi}}}}\par%
        \vspace*{-11pt}\hrulefill\\{\fontsize{9.12}{10.12}\selectfont Production and Hosted by}\par{IJRPS |~\href{https://ijrps.com}{https://ijrps.com}}\par%
        \vspace*{-2pt}{\fontsize{9.12}{10.12}\selectfont\textcopyright\ \@copyrightYear\ $|$ All rights reserved.}\par%
        \vspace*{-11pt}\rule{\linewidth}{1.2pt}
        \makeatother
        
    
\section{Introduction}
\textbf{Osteomyelitis}

Osteomyelitis (OM) is defined as an infection caused by the bone.\ensuremath{^{}}Symptoms may include pain in a specific bone area with certain indications such as redness, fever, and weakness \unskip~\citep{591514:13734267}. The disease might occur in the long bones of the arms and legs for the children, while in adults, it majorly occurs in feet, spine, and hips \unskip~\citep{591514:13734268}.

\textbf{Causes for the disease} \textit{\textbf{\space }}

OM mainly caused due to bacterial infection; generally, it is called as a fungal infection \unskip~\citep{591514:13734267}. It also may occur due to spreading from the blood or surrounding tissue \unskip~\citep{591514:13734268}. Diabetes, intravenous administration of drugs, prior removal of the spleen and trauma to a particular area are a major risk to develop OM. Diagnosis of OM can be suspected typically based on symptoms \unskip~\citep{591514:13734287}. This is then supported by blood tests, medical imaging, or bone biopsy. Depiction of a healthy bone and osteomyelitis defected bone is shown in Figure~\ref{f-b609cc742fdf}. 

\textit{Staphylococcus aureus} is the bacterial organism which is responsible for the formation of osteomyelitis. Bloodstream-sourced osteomyelitis is also caused by the above microbial agent, which is most frequently seen in children. In infants, Group B streptococci (most common microorganism; \textit{S. aureus})\unskip~\citep{591514:13734269} are seen, and \textit{Escherichia coli} are commonly isolated. In children at the age group of 1 to 16 years, \textit{S. aureus}, \textit{S. pyogenes}, and \textit{Haemophilus influenzae} are usually seen for the cause of osteomyelitis. Gram-negative bacterias, including enteric bacteria, are the significant pathogens which are the major source for the disease in the patients who are subjected to intravenous administration of the drug and also to the splenectomized patients.


\bgroup
\fixFloatSize{images/e4a01d7a-ef56-4125-afb3-dc31ee52b70d-upicture1.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/e4a01d7a-ef56-4125-afb3-dc31ee52b70d-upicture1.png}{\includegraphics{images/e4a01d7a-ef56-4125-afb3-dc31ee52b70d-upicture1.png}}{}
\makeatother 
\caption{\boldmath {Depiction of two types of bones, (a) Healthy bone and (b) Osteomyelitis affected bone}}
\label{f-b609cc742fdf}
\end{figure}
\egroup
\textbf{Biodegradable scaffolds}

Scaffolds are three-dimensional polymer matrix composed of drug implanted into biodegradable polymer fibers, used mainly for bone tissue regeneration. Biodegradable scaffolds are generally considered as indispensable elements for engineering living tissues as they are used as temporary templates with specific mechanical and biological properties similar to the native extracellular matrix (ECM). They allow modulating cell adhesion, invasion, proliferation, and differentiation; prior to the regeneration of biologically functional tissue or natural ECM. In the case of bone regeneration, a current challenge is to conceive new process strategies to fabricate composite or hybrid scaffolds able to provide three-dimensional templates and synthetic ECM environments \unskip~\citep{591514:13734270}. In the case of bone, scaffold acts as the 3D support for tissue formation. It possesses peculiar morphological and functional properties which promote cell adhesion, differentiation and proliferation, and desirable mechanical integrity to maintain the predesigned tissue structure, non-cytotoxicity, and osteoconductivity.

\textbf{Advantages of Biodegradable Scaffolds \textit{}}


  
  \begin{enumerate}
  \item \relax The main purpose of scaffolds includes retention and deliverance of cells and biochemical factors for cell attachment and migration. \unskip~\citep{591514:13734271}.
  \item \relax They also facilitate vital cells nutrient diffusion and help in the modification of the behavior of cell phase. \unskip~\citep{591514:13734272}. Biodegradable Scaffolds serve as templates in guiding the development of new tissues.
  \item \relax These are mainly used in bone and cartilage regeneration.
  \end{enumerate}
  \textbf{Microparticles}

Microparticles are enucleated phospholipid cell fragments, of diameter between 100 and 1000 nm\unskip~\citep{591514:13734273} and also usually quantified by flow cytometry and use of antibodies markers of this cell type. The drug is gradually released on erosion and diffusion from the particles. \unskip~\citep{591514:13734274,591514:13734275} Microparticles can be further categorized into 2 types, namely, microcapsules and microspheres. Double-emulsion technique is appropriate for the efficient incorporation of water-soluble peptides, proteins, and other macromolecules. This method can be used with both synthetic and natural polymers. In this technique, polymers are dissolved in an organic solvent and emulsified into an aqueous drug solution to form a w/o emulsion. \unskip~\citep{591514:13734276} The primary emulsion is subjected then to the homogenization before addition to the aqueous solution of the polyvinyl alcohol (PVA). This results in the formation of multiple (w/o/w) dispersions. The organic phase acts as a barrier between the two aqueous compartments, avoiding the diffusion of the active material to the external aqueous phase. The emulsion is then subjected to solvent removal either by solvent evaporation or by solvent extraction. Finally, the microspheres are collected by filtration and are washed with demineralized water. Microparticles provide accurate delivery of potent drugs, reduce the concentration of drug at sites other than the target tissue and serve as effective delivery systems for insoluble (or) sparingly water-soluble active agents. Microparticles increased the relative bioavailability of drugs and show taste-masking property. \unskip~\citep{591514:13734277,591514:13734278} The microparticles have great potential in reducing the dosage frequency and toxicity of various drugs. The preparative methods are simple and can be administered into the body through a hypodermic needle.

\textbf{Vancomycin for the treatment of OM }

 Vancomycin is a well-known antibiotic that is commonly used to treat several bacterial infections. Vancomycin is generally recommended for complicated infections of the joints and bones, bloodstream, skin, Staphylococcus aureus-induced meningitis and endocarditis. Vancomycin remains the most commonly used agent for osteomyelitis despite high treatment failure and recurrence rates.\-\unskip~\citep{591514:13734279}. Use of vancomycin frequently requires one or more daily infusions, based on renal function, to achieve trough concentrations of 15{\textendash}20 mg/L required for osteomyelitis.

\textbf{Importance for the development of Biodegradable Scaffolds} \textit{\space }

The major reason for the development of Biodegradable Scaffolds is to provide the antibiotic drug to be administered into the area of abrasion caused due to the disease, with the help of biodegradable polymer which will provide the drug to sweep through the bone without having any penetration at the process of treatment. The formation of scaffolds are done in such a way that, the active drug will help to deactivate or to remove the bacteria from site of the defect in the bone and the polymer will help the drug to provide sustained release and will be attached to the bone in such a way that, it will form as a bone glue at the site of abrasion and will fulfill the cavity forms due to the disease. Scaffolds are actually the polymer matrix, which will attach to the bone and will repair and protect the defected bone for a long term and the polymer matrix will get solidify according to the body temperature.

 In this study, the scaffolds are prepared with the help of Chitosan derivative polymer, which is biodegradable and also bio compatible in nature and provides the sustained release of the antibiotic drug and help in the treatment of the osteomyelitis.
    
\section{Materials and Methods}
\textbf{Materials }

Vancomycin hydrochloride drug and Chitosan polymer was purchased from Antila Lifesciences Pvt.Ltd., Ahmedabad, India. Hydroxy-proply-methyl cellulose(HPMC), methylene chloride, sodium chloride, ethanol, hydrochloric acid, methanol and glacial acetic acid was purchased from Merck Life science Pvt.Ltd., Mumbai, India. Polyethylene Glycol 600 (PEG) was purchased from Ranbaxy Laboratory Ltd., S.A.S Nagar, India. Sodium Alginate (Sodium Polymannuronate) was purchased from Loba Chemie Pvt. Ltd., Mumbai, India. Poloxamer 407 (PLURONIC\ensuremath{^{{\textregistered}}} F-127) was purchased from Sigma Aldrich, St. Louis, US. 

\textbf{Methods}

\textbf{Analytical Method}

\textbf{Standard calibration curve of Vancomycin hydrochloride in 0.1N HCl: Preparation for Calibration Curve}

Calibration curve of Vancomycin hydrochloride was prepared by taking 100mg of Drug and it dissolved in 100 ml phosphate buffer solution (pH 7.4) (PBS). The prepared solution was considered as a stock solution (Vancomycin of 1mg/ml). The stock solution was diluted to prepare a different working solution of 20, 40, 60, 80, 100, 120and 140 \ensuremath{\mu}g/ml, by taking 2, 4, 6, 8, 10, 12 and 14 mL in 100ml volumetric flask and adjusted with 0.1NHCl. The UV-Vis spectroscopy (Shimadzu-1800, Japan) was performed to determine the absorbance for an individual sample at 272nm .0.1HCl solution was taken as blank. \unskip~\citep{591514:13734280,591514:13734281}

\textbf{Formulation of Biodegradable Scaffolds} \textit{\space }

\textbf{Preparation of Vancomycin hydrochloride loaded microparticles}

In this study, Vancomycin loaded microparticles(VLM) were prepared by Double emulsion method as depicted in Figure~\ref{f-979f1eda4442}. In the first part, primary solutions were prepared by mixing 0.05g of the drug (Vancomycin hydrochloride), and specific quantities of other ingredients such as HPMC, PEG, Methylene Chloride and Sodium Alginate in a Millipore water. With the help of a glass rod, the mixture is stirred manually till a viscous solution is formed. Homogenization was carried out at the rate of 10000-15000 rpm for a duration of 5 min, with the help of Homogenizer(PT 1600 E Polytron, USA). The secondary solutions were also prepared by mixing two solutions. One solution contains Poloxamer 407 and another containing NaCl, which is solubilized in 5ml of Millipore water each. The emulsion was homogenized for 15 min at the rate of 10000 rpm. The secondary emulsion was kept in a beaker and stirring of the solution was carried out with the help of Magnetic stirrer, with the help of a magnetic bead into the solution, within a temperature of 60 to 70 {\textdegree}C, at 250-300 rpm speed. After this process, the primary solution was added to the secondary solution with the help of a 5ml syringe. The syringe was dipped into the primary solution, and the desired volume was taken out and mixed drop by drop to the secondary solution, under stirring.


\bgroup
\fixFloatSize{images/95e1fb1b-e85c-441f-bd72-9c4fe4bbb86f-upicture2.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/95e1fb1b-e85c-441f-bd72-9c4fe4bbb86f-upicture2.png}{\includegraphics{images/95e1fb1b-e85c-441f-bd72-9c4fe4bbb86f-upicture2.png}}{}
\makeatother 
\caption{\boldmath {Preparation of VLM by Double Emulsion method}}
\label{f-979f1eda4442}
\end{figure}
\egroup
The microparticles were washed with Millipore water and kept for drying in a hot air oven at the temperature of 60{\textdegree}C. The micro-particles were stored at room temperature, avoiding any contamination, by keeping it into a cellophane pouch. 

Different formulations were prepared for the preparation of the primary solution and secondary solution taking specific quantity The quantities for secondary are given in Tables~\ref{tw-354cfe8d2667}, \ref{tw-d9b445efe334} and~\ref{tw-51d9602f1b32}. \unskip~\citep{591514:13734282}.


\begin{table*}[!htbp]
\caption{\boldmath {Preparation of Primary Solution and secondary solution} }
\label{tw-354cfe8d2667}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.1371\linewidth-2\tabcolsep}p{\dimexpr.15079999999999998\linewidth-2\tabcolsep}p{\dimexpr.1321\linewidth-2\tabcolsep}p{\dimexpr.14\linewidth-2\tabcolsep}p{\dimexpr.14\linewidth-2\tabcolsep}p{\dimexpr.14\linewidth-2\tabcolsep}p{\dimexpr.16\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Pre-Formulation & HPMC (g) & PEG (g) & Methylene Chloride (ml) & Sodium Alginate (mg) & Poloxamer 407 \mbox{}\protect\newline (g) & NaCl \mbox{}\protect\newline (g)\\
\tblmidrule 
F1 &
  0.1 &
  1.5 &
  0.5 &
  1 &
  0.5 &
  1\\
F2 &
  0.2 &
  1.5 &
  0.5 &
  1.5 &
  1 &
  1\\
F3 &
  0.3 &
  1.5 &
  0.5 &
  2 &
  1.5 &
  1\\
F4 &
  0.4 &
  1.5 &
  0.5 &
  2.5 &
  2 &
  1\\
F5 &
  0.5 &
  1.5 &
  0.5 &
  3 &
  2.5 &
  1\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}
\textbf{Preparation of Biodegradable Scaffolds }

The Biodegradable preparation scaffolds have been carried out by Solvent casting method, shown in Figure~\ref{f-693b4f7eebd9}. Briefly, 50 mg, 100 mg and 150 mg of the VLM (F1 formulation)were weighed and dispersed in 10 ml of Millipore water with the help of magnetic stirrer at the temperature of 30{\textdegree}C, individually. 50mg of Chitosan was dissolved in 10 ml of glacial acetic acid and mixed with 20 ml of Millipore water. For the development of biodegradable scaffolds, VLM solutions (containing different concentration) were poured into the chitosan solution, separately and were kept for stirring on a magnetic stirrer at 350 rpm speed at the temperature of 60{\textdegree}C for 2 Hr. Development of chitosan biodegradable scaffolds containing a different ratio of VLM has been listed in Table~\ref{tw-189c04eccae4}.


\bgroup
\fixFloatSize{images/8fa7c7ca-673f-407f-a0a2-233134f89dab-upicture3.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/8fa7c7ca-673f-407f-a0a2-233134f89dab-upicture3.png}{\includegraphics{images/8fa7c7ca-673f-407f-a0a2-233134f89dab-upicture3.png}}{}
\makeatother 
\caption{\boldmath {Preparation of VLM Biodegradable Scaffolds by Solvent Casting method}}
\label{f-693b4f7eebd9}
\end{figure}
\egroup

\begin{table}[!htbp]
\caption{\boldmath {Preparation of Biodegradable Scaffolds} }
\label{tw-d9b445efe334}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.28500000000000004\linewidth-2\tabcolsep}p{\dimexpr.27119999999999997\linewidth-2\tabcolsep}p{\dimexpr.44379999999999995\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Formulation & VLM \mbox{}\protect\newline (mg) & Chitosan \mbox{}\protect\newline (mg)\\
\tblmidrule 
BS1 &
  50 &
  50\\
BS2 &
  100 &
  50\\
BS3 &
  150 &
  50\\
\tblbottomrule 
\end{tabulary}\par 
\end{table}

\bgroup
\fixFloatSize{images/d7e5a36b-c6f7-472a-b249-d66816c84248-upicture4.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/d7e5a36b-c6f7-472a-b249-d66816c84248-upicture4.png}{\includegraphics{images/d7e5a36b-c6f7-472a-b249-d66816c84248-upicture4.png}}{}
\makeatother 
\caption{\boldmath {Prepared VLM Biodegradable scaffolds from Vancomycin loaded micro-particles, with Chitosan in glacial acetic acid}}
\label{f-455ae3e6bc44}
\end{figure}
\egroup
The above Figure~\ref{f-455ae3e6bc44} shows the final formulation of VLM Biodegradable Scaffolds, which is taken for further evaluation studies. \unskip~\citep{591514:13734283}.

\textbf{Characterization and evaluation of VLM/Biodegradable Scaffolds}

\textbf{Fourier transform infrared spectroscopy(FT-IR)}

 FT-IR spectroscopy(Shimadzu-1800, Japan) was used to determine compatibility between pure drug and the physical mixture. A physical mixture of the polymer and the drug was mixed with potassium bromide (KBr), which is anhydrous in nature, in the ratio of 1:4. Around 100mg is mixed properly by means of mortar and pestle to prepare a compressed transparent and clear KBr pellet by via KBr press(Techno Search Instruments, India) which is set to 15 tons of pressure. Individually each of the pellets was taken for scanning between the wave region of 4000 to 400 cm\ensuremath{^{-1}}. Compatibility of the polymer and the drug was investigated with measuring the IR spectra of both drug and the polymer. The physical mixture, as well as the polymer, detects the shift, appearance or disappearance of the peaks \unskip~\citep{591514:13734284}.

\textbf{Differential scanning calorimetry (DSC)}

Differential scanning calorimeter(Shimadzu DSC-60, Japan) was used to perform thermal analysis. In order to confirm the compatibility of thermal behavior of naive drug and physical mixture of the drug \& the polymer DSC is used. Around 5mg of the sample was weighed and kept in a non-hermetically sealed aluminum pans and crimped. Heating is done to the samples from 0\ensuremath{^{o}}C to 350\ensuremath{^{o}}C at the rate of 10\ensuremath{^{o}}Cmin\ensuremath{^{-1}}. During the measurement, nitrogen will be continuously purged at the flow rate of 40ml min\ensuremath{^{-1}}\unskip~\citep{591514:13734285}.

\textbf{Particle size and polydispersity index analysis}

\textbf{\space }The particle size and polydispersity index of formulations (F1, F2, F3, F4, and F5) was determined by Zeta Sizer (Malvern Instruments, UK). The samples for particle size determination were prepared by dispersing different VLM formulations in Millipore water, individually followed by stirring with the help of magnetic stirrer. The samples were weighed and dissolved in Millipore water, and size measurement was conducted at 25 {\textdegree}C.

\textbf{X-Ray Diffraction (XRD)}

The physical changes in the drug during the preparation of the formulation were evaluated by X-Ray diffraction(XRD) by using X-ray diffractometer (Bruker, AXS/8, Berlin, Germany). Using CuK (\ensuremath{\alpha } radiation, 40 mA of current and 45 kV voltage), the study was carried out for the scaffolds, and they were analyzed in the diffraction angle of 2\ensuremath{\theta } 10\ensuremath{^{o}}-80\ensuremath{^{o}}. Size taken was 0.02\ensuremath{^{o}} at 2\ensuremath{\theta }. The analysis of samples were carried out at 4\ensuremath{^{o}}/min radiation scanning speed.

\textbf{Scanning electron microscopy (SEM)}

Formulated biodegradable scaffolds were placed on brass studs, and they were coated with gold with the help of ion coater. Surface morphology was observed with a scanning electron microscope (SEM)(JSM-IT300, Japan), working at an accelerating voltage of 5{\textendash}20 kV \unskip~\citep{591514:13734286}.

\textbf{Estimation of Drug content}

1g of VLM-Biodegradable Scaffold formulation was taken into 50ml of the volumetric flask, and it is diluted with 20ml of methanol. 5ml of the solution was diluted using 25ml of methanol, and 100\% dilution was made. It was subjected to analysis by UV- visible spectrophotometer(Shimadzu-1800, Japan). Drug content was determined at 272 nm \unskip~\citep{591514:13734287}.

\textbf{Percentage Yield}

\textbf{\space }Determination of percentage yield of the VLM-biodegradable Scaffolds was carried out by weighing the scaffolds and the initial ingredients such as weight of the scaffolds, the weight of the pure drug and the weight of Chitosan polymer in Equation~(\ref{dfg-3201bd9acf4d}).
\let\saveeqnno\theequation
\let\savefrac\frac
\def\dispfrac{\displaystyle\savefrac}
\begin{eqnarray}
\let\frac\dispfrac
\gdef\theequation{1}
\let\theHequation\theequation
\label{dfg-3201bd9acf4d}
\begin{array}{@{}l}\begin{array}{l}\%yield=\\\frac{wt.\;of\;Scaffolds}{(wt.\;of\;pure\;drug\;+\;wt.\;of\;C.\;P)\;}\times100\end{array}\end{array}
\end{eqnarray}
\global\let\theequation\saveeqnno
\addtocounter{equation}{-1}\ignorespaces 
\textbf{Porosity}

\textbf{\space }Determination of porosity for the prepared formulated scaffolds was done with the means of the liquid displacement method. For easy penetration of drug via the pores of the samples, ethanol has been taken as the displacement liquid \unskip~\citep{591514:13734288}. Known weights of the sample was taken and added in a known volume of ethanol. The scaffolds were removed and added to the solvent repeatedly until there is a formation of bubbles. The total volume of ethanol after adding and removing the scaffolds samples was recorded. Percentage of porosity was obtained using Equation~(\ref{dfg-283efa5c6c94}).
\let\saveeqnno\theequation
\let\savefrac\frac
\def\dispfrac{\displaystyle\savefrac}
\begin{eqnarray}
\let\frac\dispfrac
\gdef\theequation{2}
\let\theHequation\theequation
\label{dfg-283efa5c6c94}
\begin{array}{@{}l}\begin{array}{l}Porosity\;(\%)\;=\;\\(V1\;-\;V3/V2\;-\;V3)\;\times\;100\;\end{array}\end{array}
\end{eqnarray}
\global\let\theequation\saveeqnno
\addtocounter{equation}{-1}\ignorespaces 
where \textit{V1}= sample deep in a known volume of ethanol

\textit{V2= }total volume ethanol-impregnated in sample

\textit{V3}= volume remained in the beaker after removal of the sample

\textbf{Tensile Strength}

Dumb-bell shaped of the composite scaffolds (100 \ensuremath{\times} 16mm\ensuremath{^{2}}) were taken, and load-elongation measurement has been done at 5mm/min crosshead speed, at a temperature of 25\ensuremath{^{o}}Cand relative humidity of 65\%, correspondingly. Tensile testing machine (SIMPLETECH Instruments)was used for the measurement of the percentage of the elongation at the breaking point.

\textbf{In vitro drug release study}

Release of Vancomycin from the VLM biodegradable scaffolds was determined by introducing the scaffolds into a model setup of Franz-type diffusion cell at the temperature of 37\ensuremath{\pm}0.5. The formulations BS1, BS2, and BS3, as shown in Table~\ref{tw-189c04eccae4}, has been taken for the release study. PBS(pH 7.4)was kept in the receiver compartment of the apparatus. At specific intervals (0,1,2,4,6,8,12,16 and 24 Hr), 1ml of buffer solution was taken out from the receiver compartment, and the same quantity was added to maintain the volume. The whole process was carried out in a magnetic stirrer. The Vancomycin content was determined by measuring the absorbance via UV-Vis spectrophotometry technique at 272nm (Shimadzu UV 1800 ver. 2.43). With the help of Equation~(\ref{dfg-338e57e1b27e}), the percentage release of vancomycin from the scaffolds was determined using was determined.
\let\saveeqnno\theequation
\let\savefrac\frac
\def\dispfrac{\displaystyle\savefrac}
\begin{eqnarray}
\let\frac\dispfrac
\gdef\theequation{3}
\let\theHequation\theequation
\label{dfg-338e57e1b27e}
\begin{array}{@{}l}\%E\;=\;Qp/Qt\times\;100\end{array}
\end{eqnarray}
\global\let\theequation\saveeqnno
\addtocounter{equation}{-1}\ignorespaces 
Where \textit{E }is the percentage of drug release from VLM, \textit{Q}p the quantity of drug release and \textit{Q}t the total quantity of Vancomycin loaded in the scaffold \unskip~\citep{591514:13734288}.
    
\section{Results and Discussion}
\textbf{Standard Calibration Curve}

The standard plot data for Vancomycin drug in 0.1N HCl is shown in Table~\ref{tw-7174da437736}, and the Calibration curve is shown in Figure~\ref{f-e9fde69a229b}. The y value was found to be 0.006.


\begin{table}[!htbp]
\caption{\boldmath {Standard Plot data for the Vancomycin } }
\label{tw-51d9602f1b32}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.16389999999999997\linewidth-2\tabcolsep}p{\dimexpr.3594\linewidth-2\tabcolsep}p{\dimexpr.4767\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Serial No. & Concentration (\ensuremath{\mu}m/ml) & Abs at 272nm\\
\tblmidrule 
1 &
  0 &
  0\\
2 &
  20 &
  0.135\ensuremath{\pm}0.246\\
3 &
  40 &
  0.261\ensuremath{\pm}0.362\\
4 &
  60 &
  0.391\ensuremath{\pm}0.430\\
5 &
  80 &
  0.514\ensuremath{\pm}0.660\\
6 &
  100 &
  0.632\ensuremath{\pm}0.721\\
7 &
  120 &
  0.756\ensuremath{\pm}0.111\\
8 &
  140 &
  0.869\ensuremath{\pm}0.114\\
\tblbottomrule 
\end{tabulary}\par 
\end{table}

\bgroup
\fixFloatSize{images/089d3f71-1bc6-4efc-b8e6-d1ad62ebd95d-upicture5.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/089d3f71-1bc6-4efc-b8e6-d1ad62ebd95d-upicture5.png}{\includegraphics{images/089d3f71-1bc6-4efc-b8e6-d1ad62ebd95d-upicture5.png}}{}
\makeatother 
\caption{\boldmath {Standard Calibration curve of Vancomycin in 0.1 N HCl}}
\label{f-e9fde69a229b}
\end{figure}
\egroup
\textbf{FT-IR Analysis}

\textbf{FT-IR of Vancomycin Pure Drug}

Figures~\ref{f-6292dfb0dac4} and~\ref{f-4ed81f16afec} shows the FTIR spectra of Vancomycin hydrochloride drug. The spectra show broadband at 3460.45 cm\ensuremath{^{\ensuremath{-}1}}, 3250.18 cm\ensuremath{^{\ensuremath{-}1}}, 3016.23 cm\ensuremath{^{\ensuremath{-}1}}, 1663.05 cm\ensuremath{^{\ensuremath{-}1}}. The data from the study co-relates with the peaks of standard Vancomycin hydrochloride drug, as mentioned in Table~\ref{tw-15ba285e0420}.


\begin{table}[!htbp]
\caption{\boldmath {FT-IR spectra data of pure Vancomycin hydrochloride drug} }
\label{tw-189c04eccae4}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.3006\linewidth-2\tabcolsep}p{\dimexpr.43209999999999994\linewidth-2\tabcolsep}p{\dimexpr.2673\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Functional group & Standard  \mbox{}\protect\newline wave number (cm\ensuremath{^{-1}}) & The peak observed in the Vancomycin Hydrochloride drug (cm\ensuremath{^{-1}})\\
\tblmidrule 
N-H Stretching &
  3400 &
  3460.45\\
O-H Stretching &
  3200 &
  3250.18\\
C-H Stretching &
  3000 &
  3016.23\\
C=O &
  1650 &
  1663.05\\
\tblbottomrule 
\end{tabulary}\par 
\end{table}

\bgroup
\fixFloatSize{images/5b293efc-4e33-43eb-8fdf-060244b54c89-upicture6.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/5b293efc-4e33-43eb-8fdf-060244b54c89-upicture6.png}{\includegraphics{images/5b293efc-4e33-43eb-8fdf-060244b54c89-upicture6.png}}{}
\makeatother 
\caption{\boldmath {FTIR spectra of the Vancomycin hydrochloride drug}}
\label{f-6292dfb0dac4}
\end{figure}
\egroup
\textbf{FT-IR of Vancomycin and the Chitosan polymer mixture} \textit{\space }

 FTIR spectra of VLM Biodegradable Scaffolds is depicted in Figure~\ref{f-b4657c83d946}. The spectra show broadband at 3504.22 cm\ensuremath{^{-1}}, 3200.22 cm\ensuremath{^{-1}}, 3000.19 cm\ensuremath{^{-1}} and 1799.55 cm\ensuremath{^{-1}}. The data from the study co-relates with the peaks of standard Chitosan mixed with Vancomycin drug, as mentioned in Table~\ref{tw-0c10c81ca864}.


\begin{table}[!htbp]
\caption{\boldmath {FT-IR spectra data of Vancomycin and Chitosan polymer mixture} }
\label{tw-7174da437736}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.304\linewidth-2\tabcolsep}p{\dimexpr.26260000000000005\linewidth-2\tabcolsep}p{\dimexpr.4334\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Functional Group & Standard  \mbox{}\protect\newline wave number (cm\ensuremath{^{-1}}) &  The peak observed in VLM Biodegradable Scaffolds (cm\ensuremath{^{-1}})\\
\tblmidrule 
N-H Stretching &
  3500 &
  3504.22\\
O-H Stretching &
  3200 &
  3200.22\\
Ar-H Stretching &
  3000 &
  3000.19\\
C=O Stretching &
  1750 &
  1799.55\\
\tblbottomrule 
\end{tabulary}\par 
\end{table}

\bgroup
\fixFloatSize{images/cbaaefa1-82cb-40f8-b16c-ae8187d5e2ee-upicture7.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/cbaaefa1-82cb-40f8-b16c-ae8187d5e2ee-upicture7.png}{\includegraphics{images/cbaaefa1-82cb-40f8-b16c-ae8187d5e2ee-upicture7.png}}{}
\makeatother 
\caption{\boldmath {FTIR spectra of Vancomycin drug and Chitosan Polymer mixture}}
\label{f-4ed81f16afec}
\end{figure}
\egroup
\textbf{Differential Scanning Calorimetry (DSC)}

DSC thermograms of the Vancomycin drug and the physical mixture of Vancomycin drug and the Chitosan Polymer are given in Figures~\ref{f-b50236d1b77f} and~\ref{f-43d66a9b5df4}\textbf{\space }respectively.

The DSC graph shows that the optimum melting point of the Vancomycin drug is between 150-200{\textdegree}C, where in the below graph shows that, most of the vancomycin drug melted around 175{\textdegree}C. 


\bgroup
\fixFloatSize{images/c7d536a0-8ffe-4d00-9fc3-a8b18a14ed58-upicture8.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/c7d536a0-8ffe-4d00-9fc3-a8b18a14ed58-upicture8.png}{\includegraphics{images/c7d536a0-8ffe-4d00-9fc3-a8b18a14ed58-upicture8.png}}{}
\makeatother 
\caption{\boldmath {DSC curve of the Vancomycin Drug}}
\label{f-b4657c83d946}
\end{figure}
\egroup
In the DSC Graph, T\ensuremath{_{9\ }}(glass transition) value should be found around 135{\textdegree}C, according to the literature. In Figures~\ref{f-43d66a9b5df4} and~\ref{f-5277864e0e0a}, the DSC graph shows shifts in the melting point of the Vancomycin Drug to 250{\textdegree}C, due to the presence of the Chitosan Polymer used.


\bgroup
\fixFloatSize{images/1afe03ab-52d4-4e47-9f3a-817024164a21-upicture9.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/1afe03ab-52d4-4e47-9f3a-817024164a21-upicture9.png}{\includegraphics{images/1afe03ab-52d4-4e47-9f3a-817024164a21-upicture9.png}}{}
\makeatother 
\caption{\boldmath {DSC curve of Vancomycin drug and the Chitosan Polymer mixture Particle Size and Polydispersity Index(PDI) analysis of VLM}}
\label{f-b50236d1b77f}
\end{figure}
\egroup
Mean particle size and PDI of the VLM formulation (F1) was found 1624 nm and 0.245, correspondingly. Also, lower PDI of F1 revealed uniformity in particle distribution. Similarly, mean size and PDI of the other formulations (F2, F3, F4 and F5) were found as 853.9 nm with the PDI of 0.877, 1051 nm with the PDI of 0.610, 587.3 nm with the PDI of 0.722 and 1501 nm with the PDI of 0.838, respectively shown in Table~\ref{tw-00647944c24f}. The results of particles size of all the formulations have been observed that the F1 formulation was shown as higher particle size in micrometer range with lower PDI value compared to others.


\begin{table}[!htbp]
\caption{\boldmath {Cumulative results for the Particle size and PDI of the prepared VLM formulations} }
\label{tw-15ba285e0420}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.20369999999999997\linewidth-2\tabcolsep}p{\dimexpr.31620000000000005\linewidth-2\tabcolsep}p{\dimexpr.4801\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor} & Size(nm) & PDI\\
\tblmidrule 
F1 &
  1624 &
  0.254\\
F2 &
  853.9 &
  0.877\\
F3 &
  1051 &
  0.610\\
F4 &
  587.3 &
  0.722\\
F5 &
  1501 &
  0.838\\
\tblbottomrule 
\end{tabulary}\par 
\end{table}

\bgroup
\fixFloatSize{images/08413769-3025-455a-8a55-6d72c2d89401-upicture10.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/08413769-3025-455a-8a55-6d72c2d89401-upicture10.png}{\includegraphics{images/08413769-3025-455a-8a55-6d72c2d89401-upicture10.png}}{}
\makeatother 
\caption{\boldmath {Particle Size analysis of the VLM formulation F1 }}
\label{f-43d66a9b5df4}
\end{figure}
\egroup
\textbf{Drug Content} \textit{\space }

The prepared Vancomycin hydrochloride loaded microparticles(VLM) was analyzed for drug content, and the data is shown in Table~\ref{tw-0c10c81ca864}.  The obtained data of drug content formulations were ranging from 94.45\ensuremath{\pm}0.062 to 98.15\ensuremath{\pm}0.051. The highest drug content was found in Formulation F1 about 98.15 \%, and the lowest drug content was found in Formulation F4 94.45\%.


\begin{table}[!htbp]
\caption{\boldmath {Drug Content data of the Prepared VLM} }
\label{tw-0c10c81ca864}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.3184\linewidth-2\tabcolsep}p{\dimexpr.6816\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Formulations & \% Drug content\\
\tblmidrule 
F1 &
  98.15\ensuremath{\pm}0.051\\
F2 &
  95.72\ensuremath{\pm}0.045\\
F3 &
  94.80\ensuremath{\pm}0.056\\
F4 &
  94.45\ensuremath{\pm}0.062\\
F5 &
  96.88\ensuremath{\pm}0.066\\
\tblbottomrule 
\end{tabulary}\par 
\end{table}
\textbf{Percentage Yield}

The prepared Vancomycin hydrochloride loaded microparticles(VLM) was analyzed for Percentage yield. The obtained data of Percentage yield of formulations were ranging from 52.23\ensuremath{\pm}0.042 to 68.45\ensuremath{\pm}0.056. The highest drug content was found in Formulation F1 about 68.45\%, and the lowest drug content was found in Formulation F4 52.23\%. The data is shown in Table~\ref{tw-00647944c24f}.


\begin{table}[!htbp]
\caption{\boldmath {Percentage yield data of the prepared VLM} }
\label{tw-00647944c24f}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.301\linewidth-2\tabcolsep}p{\dimexpr.699\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Formulations & \% Yield \\
\tblmidrule 
F1 &
  68.45\ensuremath{\pm}0.056\\
F2 &
  66.45\ensuremath{\pm}0.051\\
F3 &
  62.18\ensuremath{\pm}0.048\\
F4 &
  52.23\ensuremath{\pm}0.042\\
F5 &
  59.48\ensuremath{\pm}0.054\\
\tblbottomrule 
\end{tabulary}\par 
\end{table}
\textbf{Scanning Electron Microscopy (SEM)}

SEM images of VLM biodegradable scaffolds (BS2) were observed at 10\ensuremath{\mu}m and 3\ensuremath{\mu}m magnificence, respectively shown in Figures~\ref{f-bb48bd7199ce} and~\ref{f-4cda71446426}. The images showed distinguish properties of the Vancomycin loaded formulation with the mixture of the Chitosan Polymer. The red highlighted circles in the images suggested that the Chitosan Polymer has coated over the VLM and the yellow highlighted circles showed the crystalline structure of the VLM Biodegradable Scaffolds.


\bgroup
\fixFloatSize{images/bfc80009-c8d1-4dd1-8b26-0bb2a2f7147c-upicture11.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/bfc80009-c8d1-4dd1-8b26-0bb2a2f7147c-upicture11.png}{\includegraphics{images/bfc80009-c8d1-4dd1-8b26-0bb2a2f7147c-upicture11.png}}{}
\makeatother 
\caption{\boldmath {SEM image of VLM Biodegradable Scaffolds at a magnification of 10\ensuremath{\mu}m}}
\label{f-5277864e0e0a}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/bde807e4-0a0c-480c-887c-630c4fd421d1-upicture12.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/bde807e4-0a0c-480c-887c-630c4fd421d1-upicture12.png}{\includegraphics{images/bde807e4-0a0c-480c-887c-630c4fd421d1-upicture12.png}}{}
\makeatother 
\caption{\boldmath {SEM image of VLM Biodegradable Scaffolds at a magnification of 3\ensuremath{\mu}m. Powder X-Ray Diffraction(PXRD)}}
\label{f-bb48bd7199ce}
\end{figure}
\egroup
PXRD analysis was performed to evaluate the crystallinity of the formulation shown in Figure~\ref{f-b7a114ec05fe}. The PXRD result of biodegradable scaffold BS2 suggested that the crystallinity nature of the powder of scaffold of chitosan with the modification with VLM formulation F1. All major characteristics peaks appear the crystallinity in the VLM biodegradable formulation, while the sharp peak at 22.9\r{6} and 31.7\r{7} 2\ensuremath{\theta } showed the presence of the vancomycin in the formulation BS2.


\bgroup
\fixFloatSize{images/366f87ba-6f4f-44c4-b6db-5f8c0d255819-upicture13.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/366f87ba-6f4f-44c4-b6db-5f8c0d255819-upicture13.png}{\includegraphics{images/366f87ba-6f4f-44c4-b6db-5f8c0d255819-upicture13.png}}{}
\makeatother 
\caption{\boldmath {XRD of the prepared VLM Biodegradable Scaffolds}}
\label{f-4cda71446426}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/56a81130-7a77-4bed-b017-6b503e462a00-upicture14.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/56a81130-7a77-4bed-b017-6b503e462a00-upicture14.png}{\includegraphics{images/56a81130-7a77-4bed-b017-6b503e462a00-upicture14.png}}{}
\makeatother 
\caption{\boldmath {In vitro drug release from the VLM Biodegradable Scaffolds}}
\label{f-b7a114ec05fe}
\end{figure}
\egroup
\textbf{Porosity} \textit{\space }

For enhancing the collagen synthesis and migration of cells, which is required for the wound healing process, the oxygen permeability for the formulated scaffolds should be more. Porosity helps in exchanging of oxygen and nutrients. The porosity of the VLM Biodegradable scaffold was found to be 75\%, by the liquid displacement method.

\textbf{Tensile Strength}

Scaffold having good mechanical property is considered by the physician to be taken care on the wound surface. The tensile strength of Vancomycin loaded microparticles biodegradable Scaffolds was found to be 7.85\ensuremath{\pm}0.58, and the elongation break was at 18.45\ensuremath{\pm}0.74.

\textbf{In-vitro drug release study} \textit{\space }

The release study of drug Vancomycin from the Chitosan Biodegradable scaffolds is shown in Figure~\ref{f-b7a114ec05fe}.  The formulations BS1, BS2, and BS3 showed an initial burst release at 1 Hr. around 34.6\%, 43.5 \%, and 44.0 \%, respectively. VLM scaffolds attain initial burst release, which caused because of non-entrapment of the drug within the microparticles. Formulation BS2 has been showed faster release as 97.1\% at 24 Hr compare to the BS1 (90.2\%) and BS3 (92.3\%). High drug content must be required at the site of infection to inhibits the wound healing, and the BS2 formulation has been suggested that approximately 97.1 \% drug release.
    
\section{Conclusion}
In this present study reveals that Biodegradable Scaffolds were successfully prepared by Solvent Casting Method. Chitosan has been taken as a suitable polymer for its higher biocompatibility and its biodegradability. Pre-formulation studies were carried out in order to establish the compatibility between the drug and the polymer by FT-IR spectroscopy. The study also revealed that the drug and the polymer were compatible. Developed VLM formulation F1 was highly effective with the mean particle size 1624 nm. Additionally, the electron micrographs (SEM) showed that the VLM formulation F1 was covered by the chitosan polymer as a biodegradable scaffold. Biodegradable scaffold BS2 has been suggested higher \textit{in-vitro} drug release as 97.1\% over the period of 24 Hr. The PXRD result of BS2 formulation exhibited crystalline nature of our developed final formulation. The drug content of the prepared biodegradable scaffolds was found to be 94.45 to 98.15\%. The percentage yield of the prepared VLM was found between the range of 52.23 to 68.45. The porosity of the prepared VLM Biodegradable scaffold was found to be 75\%. The tensile strength of VLM biodegradable Scaffolds was found to be 7.85, and the elongation break was at 18.45. Hence, it can be concluded that the chitosan polymer biodegradable scaffold of vancomycin loaded microparticles could be an effective delivery system for the treatment of osteomyelitis.
    

\bibliographystyle{pharmascope_apa-custom}

\bibliography{\jobname}
\end{document}
